Author's response to reviews

Title: Choosing care homes as the least preferred place to die: a cross-national survey of public preferences in seven European countries

Authors:

Natalia Calanzani (natalia.calanzani@ed.ac.uk)
Katrien Moens (katrien.moens@kcl.ac.uk)
Joachim Cohen (Joachim.Cohen@vub.ac.be)
Irene J Higginson (irene.higginson@kcl.ac.uk)
Richard Harding (richard.harding@kcl.ac.uk)
Luc Deliens (L.deliens@vumc.nl)
Franco Toscani (f.toscani@alice.it)
Pedro L Ferreira (pedrof@fe.uc.pt)
Claudia Bausewein (Claudia.Bausewein@med.uni-muenchen.de)
Barbara A Daveson (barbara.daveson@kcl.ac.uk)
Marjolein Gysels (mhgysels@gmail.com)
Lucas Ceulemans (lucas.ceulemans@ua.ac.be)
Barbara Gomes (barbara.gomes@kcl.ac.uk)

Version: 2 Date: 8 April 2014

Author's response to reviews:

To the BMC Palliative Care Editorial Team

Re: Revision of the manuscript “Choosing care homes as the least preferred place to die: a cross-national survey of public preferences in seven European countries”

Many thanks for your email requesting alterations to the manuscript: “Choosing care homes as the least preferred place to die: a cross-national survey of public preferences in seven European countries”. As requested, we have uploaded a revised version.

We were asked to include line and page numbers; this has now been done and the document has continuous line numbers (instead of restarting on each page). We have also removed the page breaks, although at the end there are still section breaks (since the tables are in landscape format).

Please let us know if any further alterations are required.

Kind regards

Natalia Calanzani on behalf of all the authors